Author's response to reviews

Title: BMI, waist circumference at 8 and 12 years of age and FVC and FEV1 at 12 years of age; the PIAMA birth cohort study

Authors:

Marga B Bekkers (m.bekkers@hotmail.com)
Alet H Wijga (alet.wijga@rivm.nl)
Ulrike Gehring (u.gehring@uu.nl)
Gerard H Koppelman (g.h.koppelman@umcg.nl)
Johan C de Jongste (J.c.dejongste@erasmusmc.nl)
Henriette A Smit (H.a.smit@umcutrecht.nl)
Bert Brunekreef (b.brunekreef@uu.nl)

Version: 4  Date: 28 March 2015

Author's response to reviews:

1. Move the Keywords to after the Abstract.
   ** Author's response: in the revised version the keywords have been moved to after the abstract.

2. Move the information currently placed in Acknowledgements to the Competing Interests section.
   ** Author's response: the information on the grants have been moved to the competing interests section.

3. Include details of any funding for the current study in the Acknowledgements.
   ** Author's response: the funding details of the current study have been mentioned in the acknowledgements section in the revised version of the manuscript.